Author's response to reviews

Title: Just fun or a prejudice? - Physician stereotypes in common jokes and their attribution to medical specialties by undergraduate medical students

Authors:

Martin Pyra (m.pyra@outlook.com)

Sigrid Harendza (harendza@uke.de)

Version: 2 Date: 27 Jun 2017

Author’s response to reviews:

Dear Ms Partridge,

the missing declarations "consent to publish" and "availability of data and materials" have been included.

With kind regards

Sigrid Harendza